\begin{table}%T3
%\centering
\small
   \caption{SIPS objects which share common proper motions with
   other SIPS objects. PA = Position Angle. For NLTT objects see Luyten, for LEHPM objects see Pokorny et~al. (\cite{Pokorny03}), for WT objects see Wroblewski \& Torres (\cite{Wroblewski91}) and for $^1$ see Giclas et~al. (\cite{Giclas71}). Pairs marked with~*~were identified as binaries by Luyten (\cite{Luyten88}). The pair marked with~**~was found simulatiously by Artigau et~al. (\cite{Artigau07}).}
         \label{CPMObs}
\begin{tabular}{lccrcccccl}
\hline\hline
Name&Position&{$\mu$}&PA&{$\sigma_\mu$}&{$I$}&{$J$}&{$H$}&{$K_{\rm s}$}&Other\\
&&($''$/yr)&({$^\circ$})&($''$/yr)&&&&&Name\\
\hline
\object{SIPS0650-7041}&06 50 59.75 $-$70 41 37.6&0.253&  2&0.025&16.85&14.67&14.00&13.51&\\
\object{SIPS0651-7041}&06 51 02.78 $-$70 41 38.8&0.253&  1&0.013&15.49&13.57&12.97&12.67&\\
\hline
\object{SIPS2241-5915b}&22 41 54.36 $-$59 15 30.8&0.323& 94&0.018&16.39&13.66&13.08&12.67&\\
\object{SIPS2241-5915a}&22 41 59.83 $-$59 15 12.4&0.334& 91&0.016&11.25&9.83&9.28&9.01& \object{LEHPM 5030}\\
\hline
\object{SIPS0630-5525a}&06 30 19.69 $-$55 25 48.2&0.254&137&0.007&13.71&12.42&11.87&11.59&\\
\object{SIPS0630-5525b}&06 30 23.39 $-$55 25 34.6&0.267&136&0.014&16.35&14.22&13.68&13.28&\\
\hline
\object{SIPS1917-5238b}&19 17 02.31 $-$52 38 47.0&0.254&181&0.016&11.26&9.94&9.35&9.08&\object{LDS 673b}*\\
\object{SIPS1917-5238a}&19 17 05.65 $-$52 38 49.8&0.258&182&0.014&11.40&10.19&9.64&9.31&\object{LDS 673a}*\\
\hline
\object{SIPS0358-5026}&03 58 15.41 $-$50 26 36.6&0.179& 34&0.022&16.73&14.69&14.10&13.75&\\
\object{SIPS0358-5027}&03 58 18.69 $-$50 27 39.8&0.179& 31&0.011&14.65&13.24&12.69&12.40&\\
\hline
\object{SIPS0126-5022}&01 26 55.26 $-$50 22 37.9&0.144&109&0.020&17.16&14.61&14.05&13.68&**\\
\object{SIPS0127-5023}&01 27 02.59 $-$50 23 20.2&0.170&112&0.023&17.62&14.81&14.16&13.62&**\\
\hline
\object{SIPS1131-4419}&11 31 00.17 $-$44 19 08.9&0.111&269&0.011&12.11&11.03&10.40&10.15&\\
\object{SIPS1131-4418}&11 31 01.35 $-$44 18 54.4&0.120&267&0.009&14.04&12.69&12.05&11.77&\\
\hline
\object{SIPS0207-4052b}&02 07 54.69 $-$40 52 16.4&0.255&206&0.016&12.45&11.36&10.82&10.55&\object{NLTT7110}*\\
\object{SIPS0207-4052a}&02 07 55.77 $-$40 52 30.9&0.251&209&0.015&12.53&11.49&10.96&10.66&\object{NLTT7107}*\\
\hline
\object{SIPS0116-4005b}&01 16 13.97 $-$40 05 02.1&0.170& 96&0.011&14.06&12.56&11.98&11.69&\\
\object{SIPS0116-4005a}&01 16 15.44 $-$40 05 12.1&0.157& 95&0.011&16.10&14.36&13.78&13.40&\\
\hline
\object{SIPS2145-3612}&21 45 49.39 $-$36 12 25.9&0.121&151&0.020&13.47&12.42&11.81&11.56&\\
\object{SIPS2145-3610}&21 45 54.45 $-$36 10 55.8&0.119&153&0.021&15.56&14.26&13.66&13.41&\\
\hline
\object{SIPS0252-3438a}&02 52 32.11 $-$34 38 49.0&0.308& 86&0.011&14.04&12.53&11.93&11.63&\object{LEHPM 2856}*\\
\object{SIPS0252-3438b}&02 52 33.26 $-$34 38 54.3&0.303& 86&0.011&13.69&12.25&11.68&11.38&\object{LEHPM 2857}*\\
\hline
\object{SIPS0118-2730}&01 18 10.69 $-$27 30 39.0&0.117& 90&0.011&15.86&13.90&13.27&12.95&\\
\object{SIPS0118-2729}&01 18 06.58 $-$27 29 03.3&0.115& 85&0.014&13.94&12.45&11.90&11.61&\\
\hline
\object{SIPS0215-2440}&02 15 13.69 $-$24 40 06.2&0.177& 89&0.010&14.89&13.23&12.61&12.36&\object{LDS 3363b}*\\
\object{SIPS0215-2439}&02 15 15.02 $-$24 39 43.6&0.181& 90&0.010&14.14&12.54&11.97&11.66&\object{LDS 3363a}*\\
\hline
\object{SIPS1018--2028a}&10 18 12.27 $-$20 28 22.1&0.397&285&0.010&12.21&10.59&10.01&9.71&NLTT23954*\\
\object{SIPS1018--2028b}&10 18 14.01 $-$20 28 41.9&0.407&286&0.013&10.28&9.00&8.42&8.15&NLTT23956*\\
\hline
\object{SIPS0427-1548}&04 27 20.96 $-$15 48 33.2&0.170&111&0.021&17.19&14.48&13.71&13.28&\\
\object{SIPS0427-1547}&04 27 22.20 $-$15 47 59.0&0.160&112&0.009&13.48&12.07&11.48&11.17&\\
\hline
\object{SIPS0229-1540}&02 29 44.57 $-$15 40 34.6&0.156&122&0.030&15.86&13.86&13.29&12.98&\\
\object{SIPS0229-1541}&02 29 46.98 $-$15 41 47.6&0.145&127&0.011&16.11&14.32&13.72&13.35&\\
\hline
\object{SIPS0441-1356}&04 41 58.63 $-$13 56 05.4&0.292& 86&0.028&12.63&11.43&10.87&10.60&\object{NLTT 13776}\\
\object{SIPS0442-1356}&04 42 00.44 $-$13 56 23.7&0.278& 77&0.034&15.21&12.98&12.36&11.97&\\
\hline
\object{SIPS0116-1318a}&01 16 48.12 $-$13 18 19.4&0.130&103&0.021&15.19&14.16&13.57&13.32&\\
\object{SIPS0116-1318b}&01 16 49.11 $-$13 18 55.0&0.133&105&0.021&16.42&14.93&14.41&14.07&\\
\hline
\object{SIPS1402-0312}&14 02 22.81 $-$03 12 16.9&0.401&176&0.020&12.55&11.52&10.99&10.68&\object{NLTT 36053}*\\
\object{SIPS1402-0311}&14 02 24.01 $-$03 11 55.5&0.393&175&0.021&12.20&11.08&10.57&10.28&\object{G 64-36}$^1$*\\
\hline
\object{SIPS0820-0231}&08 20 12.08 $-$02 31 09.2&0.202&166&0.031&13.24&12.13&11.60&11.29&\object{LDS 3786a}*\\
\object{SIPS0820-0230}&08 20 12.74 $-$02 30 59.5&0.170&174&0.034&13.65&12.33&11.79&11.45&\object{LDS 3786b}*\\
\hline
\object{SIPS0005-0139}&00 05 36.22 $-$01 39 39.7&0.331& 67&0.015&14.30&12.88&12.35&12.06&\object{NLTT 175}*\\
\object{SIPS0005-0139}&00 05 36.73 $-$01 39 57.2&0.336& 66&0.015&13.00&11.86&11.31&11.07&\object{NLTT 176}*\\
\hline
\object{SIPS2315-0045a}&23 15 43.90 $-$00 45 00.8&0.124& 84&0.009&14.34&12.46&11.88&11.58&\object{LDS 6019a}*\\
\object{SIPS2315-0044b}&23 15 46.52 $-$00 44 06.5&0.130& 84&0.009&14.69&12.81&12.20&11.92&\object{LDS 6019b}*\\
\hline
\end{tabular} 
\end{table}